Author’s response to reviews

Title: Cardiac patients' perceptions of neighboring patients' risk: Influence on psychological stress in the ED and subsequent posttraumatic stress

Authors:

Beatrice Konrad (bmk2145@cumc.columbia.edu)
David Hiti (dth2110@cumc.columbia.edu)
Bernard Chang (bpc2103@cumc.columbia.edu)
Jessica Retuerto (jr3448@cumc.columbia.edu)
Jacob Julian (jej2140@cumc.columbia.edu)
Donald Edmondson (dee2109@cumc.columbia.edu)

Version: 2 Date: 22 Aug 2017

Author’s response to reviews:

All the below suggested edits were made in the revised manuscript.

1. Please provide several keywords for your study at the end of Abstract.

2. Please provide a Conclusion section following Discussion.

3. Please provide list of abbreviations following Conclusion section.

4. Please cite table 2 in the main text.

5. Please provide us a clean final version without track changes.